Author's response to reviews

Title: Cognitive Testing of Physical Activity and Acculturation Questions in Recent and Long-Term Latino Immigrants

Authors:

David Berrigan (berrigad@mail.nih.gov)
Barbara Forsyth (bforsyth@umd.edu)
Cythia Helba (HELBAC@WESTAT.com)
Kerry Levin (LEVINK1@WESTAT.com)
Alicia Norberg (AliciaNorberg@westat.com)
Gordon Willis (willisg@mail.nih.gov)

Version: 3 Date: 27 July 2010

Author's response to reviews:

July 26, 2010
Dear Editor,

Thanks for the final reviews of our paper: MS: 1863820649336282.
Cognitive Testing of Physical Activity and Acculturation Questions in Recent and Long-Term Latino Immigrants in the US David Berrigan, Barbara Forsyth, Cythia Helba, Kerry Levin, Alicia Norberg and Gordon Willis.

We have submitted a revised version of the paper.

There were only two further comments.

1. We have corrected the spelling error (Seguido, not siguido)
2. The interviewers were of Mexican and Puerto Rican origin, not Central American. This point is stated on p. 14.
3. I have gone over the checklist for formatting and hope the current version matches the journals format.

Sincerely,
David Berrigan